CSCE 768 Spring 2001 Final Exam Written Monday, April 30, 2001

Do nine problems from Chapters 2-4 of Duda, Hart, and Stork’s textbook, with the following constraints:

1. no problems done in class or assigned for homework; 2. at least one problem from sections 2.1-2.3; 3. at least one problem from section 2.5; 4. at least one problem from section 2.6; 5. problem 46 in chapter 2;

6. one of problems 9, 10 and 11 in chapter 3; 7. problem 12 in chapter 3; 8. problem 35 in chapter 3;

9. two problems from chapter 4, except 12, 13, 14, 21, 22, 23, 24, 25, 26, and 30; Your problem sets are due at noon on Tuesday, May 8. A fifteen-minute oral exam based on your homework will take place at a mutually convenient time between 1 pm on May 8 and 5pm on May 11.